\begin{table}%t1
\caption{Known transiting planets by 2006$^\star$.}
%\centering
\par
\small
\label{table:planets in transit}
%\centering
\par
\begin{tabular}{cccccccccc}
\hline\hline
\# & Name &	$M_{\rm planet}$	& $R_{\rm planet}$	& Period	&	$a$ & $M_{\star}$ & $R_{\star}$ & $T_{\rm eff}$ & Metallicity\\
 & & [$ M_{\rm Jup}$] & {[$R_{\rm Jup}$]} & [day] & [AU] & [$M_\odot$] & [$R_\odot$] & [K] & [Fe/H]\\
\hline 
\multicolumn{10}{c}{OGLE planets} \\
\hline
6 & OGLE-TR-10 &	$0.63\pm 0.14$ &	$\rm 1.26_{-0.07}^{+0.07}$ &	3.10129	&	0.04162	&	$1.18 \pm 0.04$	& $1.16\pm 0.06$	& $6075\pm 86$	&	  $0.28\pm 0.10$  \\   
2&OGLE-TR-56 &	  $1.17\pm 0.04$	& $\rm 1.32_{-0.06}^{+0.06}$ &	1.211909 & 0.0225 & $1.17\pm 0.04$	& $1.32\pm 0.06$	& $6119\pm 62$	&	$0.19\pm 0.07$\\
5&OGLE-TR-111	& $0.52\pm 0.13$	& $\rm 1.067_{-0.054}^{+0.054}$	& 4.0144479	& 0.047 & $0.81\pm 0.02$	& $0.831\pm 0.031$  & $5044\pm 83$	&	$0.19\pm 0.07$\\ 
3&OGLE-TR-113	& $1.35\pm 0.22$	& $\rm 1.09_{-0.03}^{+0.03}$ & 1.4324757 & 0.0229 & $0.78\pm 0.02 $	& $0.77\pm 0.02 $	& $4804\pm 106$ & $0.15\pm 0.10$\\
4&OGLE-TR-132	&   $1.14\pm 0.12$	& $\rm 1.18_{-0.07}^{+0.07}$ & 1.689868 &	0.0299 & $1.26\pm 0.03	$ & $1.34\pm 0.08$	& $6210\pm 59$	&	$0.37\pm 0.07$\\
\hline
\multicolumn{10}{c}{Other transit survey planets} \\
\hline
7&TrES-1 & $0.76\pm 0.05$ & $\rm 1.081_{-0.029}^{+0.029}$	& 3.0300737	& 0.0393 & $0.89\pm 0.035$ &	$0.811\pm 0.020$	& $5250\pm 75$	&	$-0.02\pm 0.06$\\
11&TrES-2 & $1.28\pm 0.07$ & $\rm 1.24_{-0.06}^{+0.09}$	&	2.47063	&	0.0367 & $1.08\pm 0.08$ & $1.00\pm 0.05$	& $5960\pm 100$ & $0.15\pm 0.03$\\
10&XO-1 &	$0.90\pm 0.07$	& $\rm 1.184_{-0.018}^{+0.028}$	& 3.941634 &	0.0488 & 	 $1.0\pm 0.03$ &	$0.928\pm 0.033$ &	$5750\pm 13$	&	$0.015\pm 0.03$\\
12&HAT-P-1	&	$0.53\pm 0.04$	& $\rm 1.36_{-0.09}^{+0.011}$ & 4.46529 &	0.0551 & 	$1.12\pm 0.09$	& $1.15\pm 0.09$	& $5975\pm 45$	&	$0.13\pm 0.02$\\
13&WASP-1 &	$0.867\pm 0.073$	& $\rm 1.443_{-0.039}^{+0.039}$	& 2.519961 &	0.0382 & 	$1.15\pm 0.09$	& $1.453\pm 0.032$	& $6200\pm 200$ & \\	
14&WASP-2 & $0.88\pm 0.07$ &	$\rm 1.038_{-0.05}^{+0.05}$	& 2.152226 &	0.0307 & 	$0.79\pm 0.08$	& $0.813\pm 0.032$	& $5200\pm 200$ & \\	
\hline
\multicolumn{10}{c}{Transiting planets discovered with Radial velocities} \\
\hline
9& HD189733 &	$1.15\pm 0.04$	& $\rm 1.154_{-0.032}^{+0.032}$	& 2.218573 &	0.0313 & $0.82\pm 0.03$	& $0.758\pm 0.016$	& $5050\pm 50$	&	$-0.03\pm 0.04$\\
8&HD149026 &	$0.330\pm 0.02$ & $\rm 0.726_{-0.064}^{+0.064}$ &	2.87598 &	0.042 &	$1.3\pm 0.1$	&	$1.45\pm 0.1$ &	$6147\pm 50$	&	$0.36\pm 0.05$\\
1&HD209458 &	$0.657\pm 0.006$	& $\rm 1.320_{-0.025}^{+0.025}$	& 3.52474859 &	0.047 &	$1.09\pm 0.09$	& $1.148\pm 0.002$	& $6117\pm 26$	&	$0\pm 0.02$\\
\hline
\multicolumn{10}{l}{$M_{\rm Jup}=1.8986112\times 10^{30}~$g is the
  mass of Jupiter. $R_{\rm Jup}=71,492~$km is Jupiter's equatorial radius.}\\
\multicolumn{10}{l}{OGLE-TR-10: \citet{Bouchy_2005,Udalski_2002,Konacki_2005,Santos_2006,Pont_2006_2}.} \\
\multicolumn{10}{l}{OGLE-TR-56: \citet{Konacki_2003,Udalski_2002,Torres_2003}.} \\
\multicolumn{10}{l}{\hphantom{OGLE-TR-56: }\citet{Bouchy_2005,Santos_2006,Pont_2006_2}.} \\
\multicolumn{10}{l}{OGLE-TR-111: \citet{Pont_2004,Santos_2006,Udalski_2002,Winn_2007,Bouchy_2005}.} \\
\multicolumn{10}{l}{OGLE-TR-113: \citet{Bouchy_2004,Udalski_2002,Konacki_2004,Gillon_2006}.} \\
\multicolumn{10}{l}{OGLE-TR-132: \citet{Bouchy_2004,Udalski_2003,Moutou_2004_2,Gillon_2007}.} \\
\multicolumn{10}{l}{TRES-1: \citet{Alonso_2004,Laughlin_2005,Winn_2007}.} \\
\multicolumn{10}{l}{TRES-2: \citet{O_Donovan_2006}.} \\
\multicolumn{10}{l}{XO-1: \citet{McCullough_2006,Holman_2006,Wilson_2006}.} \\
\multicolumn{10}{l}{HAT-P-1: \citet{Bakos_2006}.} \\
\multicolumn{10}{l}{WASP-1: \citet{Collier_Cameron_2006,Shporer_2007,Charbonneau_2006}.} \\
\multicolumn{10}{l}{WASP-2: \citet{Collier_Cameron_2006,Charbonneau_2006}.} \\
\multicolumn{10}{l}{HD-189733: \citet{Bouchy_2005,Bakos_2006}.} \\
\multicolumn{10}{l}{HD-149026: \citet{Sato_2005,Charbonneau_2006}.} \\
\multicolumn{10}{l}{HD 209458: \citet{Brown_2001,Cody_2002,Wittenmyer_2005,Winn_2005,Knutson_2007}.} \smallskip \\
\multicolumn{10}{l}{\# is the label of planets in figures; they are ranked in the order of their discovery.} 
\end{tabular}
\end{table}